Author's response to reviews

Title: Potential for international spread of wild poliovirus via travellers

Authors:

Annelies Wilder-Smith (awilder-smith@ntu.edu.sg)
Wei-Yee Leong (LeongWeiYee@ntu.edu.sg)
Luis Fernandez Lopez (llopez@fiu.edu)
Marcos Amaku (amaku@usp.br)
Mikkel Quam (mikkel.quam@gmail.com)
Kamran Khan (Khank@smh.ca)
Eduardo Massad (edmassad@usp.br)

Version: 7 Date: 10 April 2015

Author's response to reviews:

Title: Potential for international spread of wild poliovirus via travellers

Responses to the Reviewers:

Thank you for the positive feedback.

Editor’s request: - add an 'Acknowledgements' paragraph in the main body of the manuscript;
- confirm whether Figure 2 was produced for this manuscript, or reproduced from elsewhere.

The “Acknowledgements' paragraph is now in the main body of the manuscript.

Figure 2 (the map) was produced by us for the purpose of this paper with the data that we calculated for this paper.

Reviewer 1 (Stephen Cochi):

665: the exact number would be 3.35 and 661.5 to make up 665. As we are talking of people here, I shortened it to 3 and 661, but I have now changed it to 3 and rounded up to 662 to make up exactly 665.

Ref 23: thank you for the correction. We have rectified it.

Reviewer 2:

Thank you for the positive comments.